Author's response to reviews

Title: Provision of relapse prevention interventions in UK NHS Stop Smoking Services: A survey

Authors:

Shade A Agboola (shade.agboola@nottingham.ac.uk)
Tim J Coleman (tim.coleman@nottingham.ac.uk)
Jo A Leonardi-Bee (jo.leonardi-bee@nottingham.ac.uk)
Andy McEwen (andy.mcewen@ucl.ac.uk)
Ann D McNeill (ann.mcneill@nottingham.ac.uk)

Version: 3  Date: 14 July 2010

Author's response to reviews:

Dear Editorial Production Team,

We have made the requested major and minor formatting changes to the paper. They are as follows:
1. The list of authors is now written as they appear in the submission system.
2. A section titled "Additional Files" has been added at the end of the manuscript.
3. Reference 4 has been amended, the reference has now been published, and this is reflected in the paper's reference list.
4. All highlighting and tracking have been removed from the manuscript.
5. The running head has been removed.
6. The visible vertical lines have been removed from the table.

Thank you.

Yours Sincerely,

Dr Agboola S.